Author's response to reviews

Title: Development of Quality Indicators for Antimicrobial Treatment in Adults with Sepsis

Authors:

Caroline M.A. van den Bosch (c.m.vandenbosch@amc.uva.nl)
Marlies E.J.L. Hulscher (Marlies.Hulscher@radboudumc.nl)
Stephanie Natsch (S.Natsch@akf.umcn.nl)
Inge Gyssens (i.gyssens@aig.umcn.nl)
Jan M. Prins (j.m.prins@amc.uva.nl)
Suzanne E. Geerlings (s.e.geerlings@amc.uva.nl)
Dutch Sepsis QI expert panel (c.m.vandenbosch@amc.uva.nl)

Version: 5
Date: 9 May 2014

Author's response to reviews: see over
To the Editor of *BMC Infectious Diseases*

Amsterdam, May 9th 2014

Re: MS: 1988538098114534

Dear Prof. Pak Leung Ho,

Thank you for accepting our manuscript "Development of quality indicators for antimicrobial treatment in adults with sepsis" for publication in *BMC Infectious Diseases*.

We added the ‘Dutch Sepsis QI expert panel’ as collaborators. I adjusted this in the main manuscript and also changed this in the submission process by adding ‘Dutch Sepsis QI expert panel’ in the ‘author details’.

Yours sincerely, on behalf of all authors,

Caroline van den Bosch, M.D.¹
Suzanne E. Geerlings, M.D. PhD ²

¹Corresponding author
²Alternate corresponding author

Academic Medical Center, University of Amsterdam
Department of Internal Medicine, Division of Infectious Diseases
Room F4-217, Meibergdreef 9,
1105 AZ Amsterdam, the Netherlands
Tel.: +31-20-5664380; Fax: +31-20-6972286
¹Email: c.m.vandenbosch@amc.nl
²Email: s.e.geerlings@amc.n